Author’s response to reviews

Title: Evaluation of lifestyle interventions to treat elevated cardiometabolic risk in primary care (E-LITE): a randomized controlled trial

Authors:

Jun Ma (maj@pamfri.org)
Abby C King (king@stanford.edu)
Sandra R Wilson (wilsoms@pamfri.org)
Lan Xiao (xiaol@pamfri.org)
Randall S Stafford (rstafford@stanford.edu)

Version: 5 Date: 5 November 2009

Author’s response to reviews:

1. On page 21, subscripted the numbers in coefficients for the expression (2) & (3).
2. For table 1, made the borderline format consistent with table 2 and changed back the bold number 7 in inclusion criteria to be consistent with other numbers.
3. Moved table 2 up one line to line up with other pages.